Author’s response to reviews

Title: Sedentary behaviour and physical activity in bronchiectasis: a cross-sectional study

Authors:

Judy M Bradley (jm.bradley@ulster.ac.uk)
Jason J Wilson (wilson-j25@email.ulster.ac.uk)
Kate Hayes (drkathrynhayes@gmail.com)
Lisa Kent (Lisa.Kent@belfasttrust.hscni.net)
Suzanne McDonough (s.mcdonough@ulster.ac.uk)
Mark A Tully (m.tully@qub.ac.uk)
Ian Bradbury (ian.bradbury2010@gmail.com)
Alison Kirk (alison.kirk@strath.ac.uk)
Denise Cosgrove (Denise.Cosgrove@belfasttrust.hscni.net)
Rory Convery (Rory.Convery@southerntrust.hscni.net)
Martin Kelly (Martin.Kelly@westerntrust.hscni.net)
J Stuart Elborn (s.elborn@qub.ac.uk)
Brenda O’Neill (b.oneill@ulster.ac.uk)

Version: 3 Date: 19 March 2015

Author’s response to reviews: see over
20/3/2015

Dear Editor

Thank you for accepting our paper entitled “Sedentary behaviour and physical activity in bronchiectasis: a cross-sectional study”. We have made the changes you have requested (see below) and uploaded the revised version of the paper.

Please let us know if we can be of further assistance.

Best Regards

Judy Bradley

1. Introduction needs to be re-named as Background.
   *This has been amended.*

2. The names of the research departments of all participating hospitals who gave approval need to be included either in the main manuscript or in an additional file.
   *These names have been included in the main manuscript under acknowledgements.*
   The authors were unsure whether to move these but are happy that journal convention is followed.

3. Ask for the funding/support section to be included in the acknowledgement section.
   *The funding was included in the acknowledgement section.*

4. Ask for the Figure legends to be moved to between the References and the tables.
   *The figure legends have been moved to between the references and the tables*